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• Completed special running with horn current 
variations, returned to default conditons.

• Projecting early September for spare target ready.
« He flow rate on current target stable.

• Repaired power supply reset circuit (~1hour).
• BPM crate reset problem on Saturday caused 

approximately 1 hour of downtime.
• Insignificant detector downtime.
• We are directly affected by the performance of the 

booster and hope its current problems can be 
quickly resolved.
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Protons Delivered in Last 
Week
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Detector Performance

Brief DAQ problem at Far Detector

Total Delivered protons 3.2 E19
Approximately 1.5 E19 per month
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Daily Protons Delivered to NuMI Target Starting 1 May 2005
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Cycle time and Beam 
Power


